Author’s response to reviews

Title: Factors related to health-related quality of life in older people with multimorbidity and high health care consumption over a two-year period

Authors:

Leonie Klompstra (leonie.klompstra@liu.se)
Anne Ekdahl (anne.ekdahl@lio.se)
Barbro Krevers (barbro.krevers@liu.se)
Anna Milberg (anna.milberg@liu.se)
Jeanette Eckerblad (Jeanette.eckerblad@ki.se)

Version: 2 Date: 18 Jun 2019

Author’s response to reviews:

1. Thank you for including information regarding ethics approval. In the Ethics approval section of the Declarations, please include the full name of the ethics committee that approved your study.

We added this information.

Line 111: Regional Ethics Committee in Linköping, Sweden (Dnr. 2011/41-31)

2. Please also remove the statement including ethics approval from the Consent to publish section, as this should only include information about consent from patients to publish their data.

We removed this information.

3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

We added this information to the manuscript.

Line 308-309: The funding bodies had no role in the design of the study; collection, analysis, interpretation of data and in writing the manuscript.
4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done.